Author’s response to reviews

Title: Using television shows to teach communication skills in internal medicine residency

Authors:

Roger Y Wong (rymwong@interchange.ubc.ca)
Sadra S Saber (sadra@interchange.ubc.ca)
Irene Ma (ima@interchange.ubc.ca)
J Mark Roberts (james.roberts@vch.ca)

Version: 5 Date: 31 January 2009

Author's response to reviews:

further formatting changes made as per request, thank you